Author’s response to reviews

Title: A QUALITATIVE STUDY OF DEVELOPING BELIEFS ABOUT HEALTH, ILLNESS AND HEALTHCARE IN MIGRANT AFRICAN WOMEN WITH GESTATIONAL DIABETES LIVING IN SWEDEN

Authors:

Katarina Hjelm (katarina.hjelm@liu.se)

Karin Bard (karin.bard@telia.com)

Jan Apelqvist (jan.apelqvist@skane.se)

Version: 2 Date: 20 Dec 2017

Author’s response to reviews:

To the Editor of
BMC Women’s Health

Dear Editor,

First we would like to express our sincere gratitude for the most valuable comments given on our manuscript entitled “A QUALITATIVE STUDY OF DEVELOPING BELIEFS ABOUT HEALTH, ILLNESS AND HEALTHCARE IN MIGRANT AFRICAN WOMEN WITH GESTATIONAL DIABETES LIVING IN SWEDEN” ref no BMWH-D-17-00042 R1, by authors Katarina Hjelm, SRN, SRNT, MScN, PhD, Professor, Karin Bard, SRN, SRNT, MScN, Jan Apelqvist, MD, PhD, Associate Professor. According to the suggestions, changes have been made and they are reported below.

Enclosed please find a report on changes made and a clean version of the revised manuscript.

Sincerely yours,

Katarina Hjelm, SRNT, MSc, PhD, Professor

Department of Social and Welfare Studies,
Linkoping University,
Campus Norrkoping
S- 601 74 Norrköping
Sweden

e-mail: katarina.hjelm@liu.se
First we would like to express our sincere gratitude for the most valuable comments given on our manuscript. According to the suggestions from the editor, changes have been made and a file with the clean version of the manuscript without any changes made marked is attached.

1) The “Contribution of the Paper” section has been removed and the text is incorporated in the manuscript.

2) A statement at the end of the section Authors’ Contributions is included to confirm that all authors have read and approved the final version of the manuscript.

3) The content of the interview guide is stated under Data collection on page 12 in the manuscript.

4) A single, clean copy of the manuscript, without tracked changes or highlights, is provided as an attachment.